EFO Group Meeting	7/23/2019
Schmidt, White, Vanwingeren, Mazzio,  Flood, Berkland, Wyncott, Moonier, Pedergnana, Rossi
Fieramosca-Off

Patti-6-ID. CCRT this Thursday.  They are looking to do SV of D Extension and beam transport this Friday.
2-ID CANT Project to start in shutdown,  will let us know when RSS tags should be removed.  Look for WRs
25-IDA Caratelli to start in November 2019
28-ID Utlities- Electric is out to bed 

Liz- Sector 10ID High pressure experiment.  Set up was inspected by Ken Volin and Rossi.  Some minor follow ups and electrical inspection but Experiment will be lowered to medium risk and approved with comments.
Tarantulas are come to sector 18 soon
User who hand carried vials of samples is coming back to sector 12.  Nena will address the user to ensure they know to ship samples
Sector 3 walkthrough was cancelled and moved to Wednesday
Sector 13 walkthrough completed
Clay:  Gentle reminders mentioned below.
Wendy- Went over highlights from Shutdown planning meeting.  See the email she sent. Work requests are coming in, please approve before Wednesday Shutdown meeting
Rad-Tuesday set up for 9-IDC and 16-BMD and another set up for Friday.
Paul-Stockroom inventory on disposable gloves- all expired boxes where removed.  As you go around check your areas.  Also Shane check our supply of cryo gloves.  Paul needs follow up on a few items listed below.
John- Sector 13 walkthrough completed today. No outstanding issues.
Several Routine Rads this week.   Tuesday 23-ID and Wednesday 10-IDB and 3-ID
20-BM shutter warnings will be addressed during the shutdown as it is not hampering the ability to get beam 
Lisa- On shift this week.   Soil permit transfer to Liz is underway.  Lisa requested amendment to the current permit to have it under Liz’s name.   Lisa and John to go over 434 activities before she transitions to the MCR.
Shane- Discussed the two ongoing issues.  32-ID storage access today to change PS2 switch after repeated trips over the weekend.  Beamline Downtime started and needs to be completed
[bookmark: _GoBack]23-IDB warnings that prohibited search and secure to be completed.   SI group investigating the problem.
Gentle Reminder:  
Check your area ESAFs in the table vs the beamline cabinets.  Please be diligent about deactivating ESAFs that can be and removing them from the cabinet and table.
Also after any studies please make sure you reset PSS faults/trips that occurred from work in your area before the start of User Beam.   This will eliminate extra calls during a busy time.
Work Requests are coming in for shutdown work.  Please approve your areas as they come in.
Action Items:
Wendy and Dean-Follow up with 6-ID and if SV is planned for Friday ensure things are in place during Wendy's absence on Friday.
Liz- Follow up with 10-ID high pressure ESAF and approve when inspections have been completed
Clay and John-Follow up on Paul Rossi's request for compressed gases in the 434/435 truck lot.  Please talk to beamlines and determine if oxidizer cabinet is needed and where it can be placed
ALL FCs-  Please give Wendy the drop locations for switchgear maintenance.  Wendy needs to compile this list for Ron Moore so MOM group has an idea what will be needed.   We need estimations.
John- Check out the disposable gloves in 434 areas specifically 14-BMC.  John will check all the labs in 434/435.
Dean-To add the welding points in the 6-ID CCCL list
Nena-Give an overview of Component Database and e traveller after her meeting with Betsy Dunn
Meetings
Several meetings this week.  Here are the list and who agreed (coerced) to attend.  Please report back information so we can all stay informed.
Wednesday July 24 9am Shutdown Planning Meeting- Wendy
Wednesday July 24    11 am 28-ID Utilities meeting-Shane
Wednesday July 24 1pm. All Hands Meeting-Anyone.
Wednesday July 24 3pm Sector 3 Safety Walkthrough- Bruno and Liz
Thursday July 25 9am  2-ID Cant Project-Dean
Thursday July 25 1pm CCRT for 6-ID- Wendy/Dean(in case there is a SV on Friday)


